Author's response to reviews

Title: Decentralization of the provision of health services to people living with HIV/AIDS in rural China: The case of three counties

Authors:

Xiulan Zhang (zhang99@bnu.edu.cn)
Pierre Miege (miege@hotmail.com)
Yurong Zhang (yurongbnu@gmail.com)
Xiaohua Wang (wxhbnu@126.com)
Qiujie Zhang (zhangqiujie@126.com)

Version: 2 Date: 24 November 2010

Author's response to reviews:

Dear Editor,

I am enclosing herewith a revised manuscript, now entitled “Decentralization of the Provision of Health Services to People Living with HIV/AIDS in Rural China: The Case of Three Counties HIV/AIDS Health Services”.

With the submission of this manuscript I would like to undertake that the changes have been made following the comments provided by the three reviewers. The tables were revised in order to make the data collected clearer. The legislative framework and main policies concerning treatment and care of HIV/AIDS patient have been added and more emphasis was given to the presentation of the policy changes in the last decade. Therefore, the specific challenges of the Chinese decentralization system have been clarified, and the importance of the research we present was highlighted. The presentation of this study’s methodology has been expanded to answer the questions of the reviewers. The paper was proofread to check for language mistakes.

Thank you for your time and consideration.

Sincerely,

Pr. Pierre Miège